

G OPEN ACCESS

Citation: Lam SD, Babu MM, Lees J, Orengo CA (2021) Biological impact of mutually exclusive exon switching. PLoS Comput Biol 17(3): e1008708. https://doi.org/10.1371/journal. pcbi.1008708

Editor: Marco Punta, San Raffaele Hospital: IRCCS Ospedale San Raffaele, ITALY

Received: July 1, 2020

Accepted: January 14, 2021

Published: March 2, 2021

Copyright: © 2021 Lam et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The authors confirm that all data underlying the findings are fully available without restriction. All relevant data are within the paper and its Supporting Information files. All MXEs can be accessed via the MXE-MOD webserver (http://gene3d.biochem.ucl.ac.uk/ mxemod/). Structural models and details of the homologous exon switches with links to the corresponding structural models are available at: ftp://orengoftp.biochem.ucl.ac.uk/gene3d/ CURRENT_RELEASE/mxe-mod/. **RESEARCH ARTICLE**

Biological impact of mutually exclusive exon switching

Su Datt Lam^{1,2}*, M. Madan Babu^{3,4}, Jonathan Lees⁵*, Christine A. Orengo¹*

1 Institute of Structural and Molecular Biology, University College London, Darwin Building, Gower Street, London, United Kingdom, 2 Department of Applied Physics, Faculty of Science and Technology, Universiti Kebangsaan Malaysia, Bangi, Malaysia, 3 MRC Laboratory of Molecular Biology, Francis Crick Avenue, Cambridge Biomedical Campus, Cambridge, United Kingdom, 4 Department of Structural Biology and Center for Data Driven Discovery, St Jude Children's Research Hospital, Memphis, Tennessee, United States of America, 5 Faculty of Health and Life Sciences, Oxford Brookes University, Oxford, United Kingdom

• These authors contributed equally to this work.

* sudatt@ukm.edu.my (SDL); jlees@brookes.ac.uk (JL); c.orengo@ucl.ac.uk (CO)

Abstract

Alternative splicing can expand the diversity of proteomes. Homologous mutually exclusive exons (MXEs) originate from the same ancestral exon and result in polypeptides with similar structural properties but altered sequence. Why would some genes switch homologous exons and what are their biological impact? Here, we analyse the extent of sequence, structural and functional variability in MXEs and report the first large scale, structure-based analysis of the biological impact of MXE events from different genomes. MXE-specific residues tend to map to single domains, are highly enriched in surface exposed residues and cluster at or near protein functional sites. Thus, MXE events are likely to maintain the protein fold, but alter specificity and selectivity of protein function. This comprehensive resource of MXE events and their annotations is available at: http://gene3d.biochem.ucl.ac.uk/mxemod/.

These findings highlight how small, but significant changes at critical positions on a protein surface are exploited in evolution to alter function.

Author summary

A single gene can produce more than one protein through processes such as Alternative Splicing. Mutually exclusive splicing is one type of splicing by which cells produce isoforms that are often very similar in their three-dimensional shape. Different transcript isoforms are usually produced in different cell tissues and cell environments. The structural impacts of this splicing are still not well-studied. In this study, we compare proteins generated from mutually exclusive exon switching from 5 organisms. Most of these proteins do not have their three-dimensional shape solved by experimental methods. We therefore calculate the most probable three-dimensional structural models. We found small regions that are different between the proteins and occur near important residue sites that determine the functions of the proteins. We speculate mutually exclusive splicing may be involved in tuning the protein function. We have built a webserver to present information on these exon switching events: http://gene3d.biochem.ucl.ac.uk/mxemod/.

Funding: This work was supported by the Biotechnology and Biological Sciences Research Council [https://bbsrc.ukri.org/, BB/L002817/1 to JL] and Fundamental Research Grant Scheme from the Ministry of Higher Education Malaysia [http://portal.mygrants.gov.my/main.php, FRGS/1/ 2020/STG01/UKM/02/3 to SDL]. MMB acknowledges UKRI Medical Research Council (https://mrc.ukri.org/, MC_U105185859) and the American Lebanese Syrian Associated Charities (ALSAC, https://www.stjude.org/about-st-jude/fag/ whats-alsac.html) for support. Funding for open access charge: Biotechnology and Biological Sciences Research Council (https://bbsrc.ukri.org/). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Introduction

Alternative splicing (AS) refers to the assembly and rearrangement of different exons of a gene during pre-mRNA splicing such that different mRNAs and thus proteins are produced from the same gene. Alternative splicing is common in humans and other animals producing different transcripts in different developmental stages, tissues or disease states [1,2]. Alternative splicing events have been linked to various diseases and cancers [3–6] and may lead to the tissue-specific rewiring of protein-protein networks [7–9]. A biologically meaningful role for many AS events has been demonstrated [10–12], but the extent to which AS extends the functional repertoire is a subject of considerable active debate in the literature [13,14].

Amongst the most important exon-switching events are those that generate homologous mutually exclusive exons (MXEs), such that only one out of the two exons is retained while the other one is always spliced out. This type of exon switching is less likely to be disruptive to highly organised globular protein structure [14,15] compared to AS events such as cassette exon removal. Consistent with this notion, homologous MXEs have been found to be highly enriched in proteomics experiments [16] and to be more conserved between species [17] compared to alternative transcripts generated by Alternative Splicing in general. MXEs have been shown to be enriched with muscle and membrane functions (such as transporter, signal transduction) [18,19].

It is possible to align a pair of homologous MXE events and identify conserved and variable amino acid residues, the latter of which could be considered responsible for any functional shifts of the MXE (Fig 1A). That is, changes in amino acids could contribute to changes in specific functions and there are several examples of this in the literature (Fig 1B). For instance, specific examples have been found for which MXE events modulate the binding of the protein to a substrate (e.g. protein, ion) [14,17] and have been shown to alter the voltage dependencies of ion channels [20].

Given that MXEs are likely to affect protein function, how often and what types of functions are altered? Here, we analyse multiple genomes and use large scale protein structure modelling data to assess the functional impact of these processes and their likely biological role. We describe the mechanisms by which MXE events trigger dynamic switching of protein surface patches associated with key solvent exposed functions (i.e. binding to other biomolecules such as proteins, small molecules, etc.). We subsequently illustrate how aberrations in the same functional switching regions altered by MXE events may be mutated leading to deregulation in cancer. Unlike MXE events, many AS events such as cassette exon removal in globular regions, are likely to lead to significant disruption of the protein structure. However, for a subset of cassette exon splicing events where there is less likely to be structural disruption, and microexons in general we do not identify the same functional enrichments as for MXE events. Finally, we present a comprehensive online resource of the structural and functional annotations of MXE events to allow users to search for MXEs of interest to investigate the structural/functional consequences of these events.

Results

Comprehensive protein structure modelling of MXE events from multiple species

Through a comprehensive genome-scale analysis pipeline, we first identified homologous MXEs in 5 high-quality Metazoan genomes (human, fly, mouse, fugu fish and zebrafish) (see Materials and Methods). In this way we could compare results across genomes to examine



Fig 1. Identification of variable residues and possible functional effects of variable residues. (A) Identifying the variable residues for an MXE splice event (MXE-specific residues). The same colour code is used throughout the paper. The MXE region from the pair of proteins generated from the splicing are shown in purple or yellow. Variable residues from the MXE event are shown in red. (B) Possible functional effects of variable residue switching assessed in this paper include altering PPI (Protein-Protein interactions), CSA (catalytic residues from the Catalytic Site Atlas[21]), PSI (Protein-Small molecule Interactions), Allosteric, PTM (Post-Translational Modification) and FunSites (predicted functional sites from functional families (FunFams)).

https://doi.org/10.1371/journal.pcbi.1008708.g001

any potential similarities and variations in the different species. We find that in our dataset, each organism has over one hundred MXE containing genes (Fig 2A, see S1 Table for list of MXE genes from each species). The MXE protein sequence pairs in our dataset typically have high sequence identity with a mean of 67% (Fig 2C) and a median length of 36 amino acids (Fig 2B). Apart from the major peak there is a second smaller peak at higher sequence similarity, which could arise from a number of factors including biases in protein evolutionary rate [22]. Functional differences in protein sequences containing the MXE will arise from the variable residues (i.e. amino acid residues that are variable in the alignment between MXE exons, Fig 1). There are usually less than 10 variable residues (median = 6 residues) (Fig 2D).

An analysis of the evolutionary conservation of MXE events revealed that 58% of human MXE genes have an ortholog in mouse that is also an MXE gene, but for other organisms the gene overlaps were considerably lower (Fig 2A). This relatively low overlap of MXE genes in our dataset, does not mean the MXEs are absent from these other organisms, since more



Fig 2. General statistics of our MXE dataset. (A) Table showing the number of MXE events and MXE genes in each organism and a matrix with the percentage overlap of MXE gene orthologs between species. The percentage overlap refers to percentages of overlap between gene orthologs between species (e.g. 58% of human MXE genes have an ortholog in mouse). (B) Distributions of MXE exon sizes. MXE sizes were calculated by calculating the length of the MXE. The length of the longer isoforms were plotted. (C) Distribution of sequence identity between MXE pairs identifies one main distribution with its main peak between 50% and 70% sequence revealing likely conserved structure and function. A second much smaller peak around 90% indicates a subset of MXE events that are more recently evolved. Sequence identifies were calculated by BLASTing two isoforms against each other. (D) Distribution of number of variable residues between MXE pairs demonstrates that the majority of MXEs have only a limited number of variable residues. Variable residues are amino acid residues that are variable in the alignment between MXE pairs.

https://doi.org/10.1371/journal.pcbi.1008708.g002

targeted studies point to general conservation of MXEs, but does illustrate that our multi-species approach considerably extends the scope of our analysis by giving many additional MXE events, to better reveal the general principles of MXEs.

CATH superfamilies are groups of homologous protein domains (i.e. having significant structural and/or sequence similarity) [23]. Each CATH superfamily can be further subdivided into functional families (called FunFams) providing more structurally and functionally coherent sets of protein domain homologues. FunFams have been used for predicting protein functions and for analysing the possible functional impacts of disease variations (i.e. residue mutations) in proteins [23,24]. We analysed the structural features of our MXE splicing events i.e. their location in the protein structure and their proximity to known and predicted functional sites, using the FunMod pipeline (derived from CATH FunFams, see <u>Materials and Methods</u> and cited references [25,26]) and we supplemented this using direct structural mapping guided by sequence alignments (see <u>Materials and Methods</u>). The results showed that whilst only a very small number of MXE events can be mapped to known structures, we can massively expand our structural coverage using our structural modelling strategy, from 14 to 691 MXE events.

Names of superfamilies common to all five species

to five species	
	1.10.51
	1.20.5.
	1.20.5.3
the state of the s	2.10.25
	2.60.40
	3.40.50
32 1 1	1.10.23
	2.10.11
Not	2.70.15
	3.40.19

CATH superfamilies common

.10.510.10Transferase (Phosphotransferase)
domain.20.5.110vesicle-mediated transport protein.20.5.340myosin/tropomyosin.10.25.10Laminin (influencing cell differentiation,
migration, and adhesion).60.40.10Immunoglobulins (antibody).40.50.300P-loop containing nucleotide
triphosphate hydrolases

Names of superfamilies common to four species, except fly

1.10.238.10	EF-hand
2.10.110.10	Cysteine Rich Protein (kinase related)
2.70.150.10	Calcium-transporting ATPase, cytoplasmic transduction domain
3.40.190.10	Periplasmic binding protein-like domain
3.40.1380.20	Pyruvate kinase

Fig 3. Number and names of CATH superfamilies common to multiple species analysed in this study. MXEs from different species were mapped to CATH domain superfamilies. CATH superfamilies are groups of protein domains with clear evidence of homology. The CATH superfamily code is denoted by four numbers corresponding to each level in the CATH classification (i.e. 3.20.20.120). At the top of the hierarchy is the class level where structural domains are classified based on their secondary structure content. The second level of the hierarchy is the architecture level given by the global arrangement of secondary structures in 3D space. This is followed by the topology level where domains with similar folds (which takes into account the 3D arrangement, orientations and connections between the secondary structures) are grouped together. The fourth level is the homologous superfamily level where domains are deemed homologous. A website tool (http://bioinformatics.psb.ugent.be/webtools/Venn/) was used to draw this Venn diagram.

https://doi.org/10.1371/journal.pcbi.1008708.g003

The MXE regions structurally mapped to a relatively small number (115 out of a possible 2700) of CATH superfamilies (Fig 3). Preliminary analysis of the CATH superfamily functions revealed that the MXE domain families are enriched in important Metazoan functions such as membrane proteins involved in cell-cell adhesion and signal transduction (see S1 Fig). Functional analysis by the PANTHER pipeline [27] showed additional functional enrichments from MXE genes, in membrane proteins (e.g. ion channels, synaptic vesicles and receptors, FDR level <0.01; See S2–S6 Figs for details). Our analyses show similar enrichments as other previous MXE studies [18,19].

Why have some proteins, such as ion channels, synaptic vesicles and receptors evolved by expanding their functional repertoire through MXE events? There could be some high level of association between these MXE containing genes that has not yet been considered. For example, one might suspect that MXE functional expansion is somehow advantageous for genes expressed in certain tissues since rewiring of tissue specific networks is thought to be a function of other types of protein isoform switching [7]. TopAnat provides a method to detect anatomical terms that are enriched in a list of genes (in terms of their expression in different anatomical regions of human tissues) [28,29]. Using the TopAnat website, we found a significant enrichment for human MXE genes to be expressed in various brain anatomies, with the highest significance for the anterior prefrontal cortex (Brodmann (1909) area 10) (FDR = $2e^{-6}$,

Fold enrichment = 2.05). Brain and neural tissue enrichments can also be found for the other classes of splicing (cassette exon and microexon, discussed in S2 and S3 Texts). Other studies have shown enrichment of AS events in neuronal tissue [30]. Here we are simply capturing the tissue gene expression enrichment (rather than the occurrence of splicing in these tissues). Our analysis, therefore, implies that MXE genes tend to have functions associated with neuronal tissues.

Finally, using a set of FDA druggable proteins from the Human Protein Atlas (HPA) [31], we found significant enrichment of druggable proteins in our human MXE genes (fisherstest = p-value 7.8e⁻⁵, odds-ratio = 3.9). In total 13 of the 672 FDA approved genes available from HPA, could be mapped to one of the 125 human MXE genes (see S2 Table and MXE website for list of druggable targets). This indicates that MXE genes are in general likely to be interesting targets for disease therapies. An example of a known drug targeting an MXE gene is a voltage gated Calcium channel CACNA1D which is targeted by a number of drugs for hypertension such as Amlodipine (DrugBank:DB00381). We saw a similar enrichment for druggable genes using another dataset of MXE events compiled by Hatje and co-workers and described in S1 Text (p-value < $1.2e^{-6}$, odds-ratio = 2.8). We also analysed single exon loss events (see Materials and Methods) and saw a significant enrichment (p-value = $9.7e^{-6}$) but with a substantially lower odds-ratio of 1.6, whilst microexons showed no enrichment.

Large-scale structural analysis using 3D structures and models shows that MXE events predominantly modulate surface residues and protein binding functions

The function of a protein domain is largely dictated by the shape and nature of the residues on the protein surface. Enrichment of MXE events on surface residues would indicate absence of an effect on the protein fold but an effect on function such as altered protein interactions, protein-small molecule binding, etc. Our structural mapping approach (see Materials and Methods) provided structural information to assess this (S7 Fig) for 691 MXE splice events. We observed that the percentage of exposed residues for an MXE region is significantly higher than what one would expect by chance (Fig 4A) (p-value = $1.9e^{-144}$, Wilcoxon signed-rank test). The level of the difference was striking and indicates that MXE events are strongly biased to altering residues with surface exposure. This picture is consistent across multiple organisms and indicates this is a prominent and consistent feature of MXE splicing.

Are the variable residues involved in MXE events more exposed to the solvent than the MXE region itself? A more detailed analysis, involving buried surface area calculations for all the MXEs revealed that variable (i.e. MXE-specific) residues are much more significantly exposed than non-variable MXE residues (Fig 4B; p-value = $2.3e^{-37}$, Wilcoxon signed-rank test). These observations further demonstrate that the amino acid changes in MXE events consistently alter surface exposed residues on the protein, much more than we would expect by chance. Together these findings collectively suggest a role for these surface variable regions in modulating protein functional sites.

How drastic are the amino acid changes in the variable sites and do such changes happen at or near functionally relevant sites on the structure? Alterations in the variable residues for an MXE event can be classed as conservative or radical using established residue substitution scoring matrices, with the latter more likely to have functional consequences. We compared the physicochemical properties of the equivalent MXE variable residues using the McLachlan physicochemical matrix [32] which captures chemical similarity between residues. Typically, McLachlan scores < 2 for an amino acid change are used to indicate a significant change in physicochemical properties [32]. For all the organism datasets, more than 90% of the MXE



Fig 4. Structural and functional analysis of the MXE splicing dataset. (A) The surface exposure of the MXE events compared to random expectation. The first quartile, median, and the third quartile of the population were indicated with lines. NACCESS was used to calculate the relative accessible surface (rASA) of amino acids. Amino acid residues were considered to be exposed if the rASA value was above 10%. (B) The surface exposure of the MXE variable residues compared to the MXE non-variable residues. We compared the solvent exposure of the variable residues versus the solvent exposure of the MXE conserved residues for each MXE event. NACCESS was used to calculate the relative accessible surface (rASA) of amino acids. Amino acid residues were considered to be exposed if the rASA value was above 10%. (C) Distribution of McLachlan scores for the variable residues. For each MXE, we summed up all the McLachlan similarity scores for a set of variable residues and divided by the total number of mapped variable residues from the Catalytic Site Atlas, PPI = protein interaction sites and PSI = protein-small molecule interactions). A breakdown of this Fig by individual species is available in \$10 Fig. We used the Z-score test to compute the statistical significance. Where appropriate the level of statistical significance is shown above each type of functional site.

https://doi.org/10.1371/journal.pcbi.1008708.g004

variable residues have a McLachlan score ≤ 2 (Fig 4C). These results indicate that in general the MXE events produce significant changes in physicochemical properties of the MXE regions. We also compared the physicochemical changes observed for the MXE variable residues using a random model which randomly samples residues within the same threshold distance of 4Å from a particular class of functional residue (catalytic, ligand binding, PPI). We find the tendency for chemical change in the MXE variable residues to be significantly different to random (all with p-values $\leq 6.35e^{-125}$, See S8 Fig). Such considerable changes suggest that MXE events are likely to cause functional shifts between protein isoforms, particularly if they lie on or close to functional sites.

Do the MXE variable residues cluster together on the protein structure (or are they dispersed) and do they tend to affect functionally relevant sites? A detailed analysis of the positioning of the MXE variable residues revealed that they tend to form a single cluster in space in over 70% of the MXE events. For those MXE events that formed multiple clusters, they usually formed 2–3 clusters (S9 Fig). Next, we determined whether these clustered variable residues map to known functional sites or lie close to them. We first identified well-established functional sites such as catalytic residues, positions that are known to mediate protein-protein and protein-small molecule interactions (PSI) based on crystal structures of available homologs within the functional family to which the MXE event has been mapped (Fig 1B). We also identified positions that are likely to confer functional selectivity using the FunSites approach (See <u>Materials and Methods</u>). Briefly, we mapped the MXE region onto CATH functional families (FunFams). Positions that are conserved within and unique to a particular FunFam but not conserved across the superfamily are likely to be specificity-determining positions and responsible for a specific function that is distinct from other superfamily members. Finally, we also identified putative allosteric sites using the A-SITE approach (see <u>Materials and Methods</u>).

Having annotated the residue positions, for every pair of splice isoform compared, we determined if the centre of mass of the cluster of MXE variable residues was close to any of the above-defined functional sites (within 6Å). We calculated the minimum distance between the atoms of the residues involved. A comparison with a random model (Materials and Methods) shows that there is a significant tendency for clusters of MXE variable residues to lie close to protein-protein interaction, protein-small molecule, specificity determining positions and allosteric sites for all species (Fig 4D) (all with p-values $\leq 3.55e^{-03}$, please refer to Fig 4D for details of these values). The signal was strongest for protein-small molecule interaction (PSI) and protein-protein interaction (PPI) functional sites. We also performed the analysis without clustering the variable residues and obtained similar results (see S11 Fig for details).

Through our structural modelling pipeline, integrated with functional site datasets, a clear picture emerges for a consistent role of MXE events modulating functional regions in domains. In line with previous work, our modelling suggests that the sequence variations in MXEs are typically small and although chemically quite extreme (as judged by the McLachlan index), they are usually surface exposed and therefore unlikely to affect the ability to produce stable structures but more likely to alter surface functions.

To confirm our observations, we performed a second independent analysis using the more comprehensive validated human MXE-splicing events from the Kassiopeia resource which we refer to as the Hatje dataset (N.B. this dataset was filtered to only include homologous exons). We found similar properties with the Hatje dataset further supporting the generality of our observations (See S1 Text and S19–S24 Figs). We also studied the structural and functional effects of MXEs with cassette exon splicing, we identified single exon loss events (see Materials and Methods) and found no enrichment of cassette exon near functional sites (i.e. PPIs, PSIs etc, See S2 Text and S25–S33 Figs). This along with enrichment in proteomics highlights an important and distinct functional role for MXEs. A sub-class of cassette exons are microexons which are small, surface exposed and therefore less likely to damage globular protein structure than cassette exons in general. We also found that they were surface exposed but did not find they were enriched in any of the functional locations we identified for MXEs (S3 Text and S34–S40 Figs).

Variable regions in MXE events lie at or close to cancer mutation sites

How often do variable regions lie at or close to cancer mutation sites? Recent studies have suggested an association between human MXE events and inherited pathogenic mutations [18]. Furthermore, many studies have found that cancer mutation sites tend to lie in the vicinity of functional residues (i.e. protein and small molecule binding sites) [33,34]. MutFams are CATH functional families found to be significantly enriched in cancer mutations (see <u>Materials and Methods</u>). We therefore investigated whether MutFam cancer mutation residues coincide with both MXE variable residues and functional residues. We mapped putative cancer driver genes from MutFams to the 143 human MXE events. We found carcinoma (e.g. adenocariconoma, squamous cell carcinoma) and melanoma to be the common cancers that have mutations coinciding with MXE events (by analysing cancer annotations in the MutFam). Functional analysis by the PANTHER pipeline [27] on cancer-related MXE showed additional functional enrichments, in signalling process (e.g. oxytocin, adipocytokine and prolactin) and regulatory pathway (e.g. insulin secretion, transcription) (FDR level <0.01; See S12 Fig for details).

We could annotate 43 of these events with structures and we then determined if the variable residues/clusters in these MXE events were close to MutFam mutation residues. We found that 40 of the 43 MXE events had variable residues that were significantly closer (in terms of distance) to cancer mutations than expected by random (p-value <0.0001, see Materials and Methods). We obtained the same results for both per-residue and per-cluster analyses. We checked if these MXE events were close to functional residues and found 35 out of 40 of the events were close to functional residues. Similarly, there is a significant tendency for cancer mutation and variable residues to lie close to protein-protein interaction, protein-small molecule, specificity determining positions and allosteric sites. Hence, MutFam cancer mutations target a similar set of functional sites as are altered by MXE events, suggesting some advantage to cancer in modifying the same set of specific functional residues on proteins that are dynamically regulated by MXE events. We found similar properties with the Hatje dataset further supporting the generality of our observations (See <u>S1 Text</u> for more information).

Molecular principles of the impact of MXE events

Results from the global analyses of MXE events using the approaches described above have suggested general principles and mechanisms by which MXE events influence protein function and how cancer mutations tend to affect MXE variable positions or fall near them, thereby influencing disease states. In this section, we present four examples that illustrate the above described findings (Further details of these examples are available in the S4 Text).

Protein interaction. A total of 260 MXE events are associated with PPI functions. For example, the PKM gene is a pyruvate kinase enzyme with an essential role in metabolism and is important in the growth of cancer cells [35]. Although other isoforms are also present, the two main PKM isoforms generated by MXE splicing (known as PKM1 and PKM2 isoforms) and the MXE event alters variable residues of its fourth domain (S13 Fig), mostly on the protein surface (S13 Fig), associated with allosteric (by FBP small molecule interaction) and protein interaction (tetramerisation) functions (Fig 5A). PKM2 dimers and tetramers possess low and high levels of enzyme activity, respectively, and this functionality (oligomerisation state switching) is important for proliferating cells (when PKM is in its inactive dimeric state, preceding pathway intermediates build up and are driven down alternative pathways to generate biomolecules important for growth). Further details on this are given in S4 Text. Our analysis (see Materials and Methods, S14 Fig and S4 Text) show that the effect of the MXE variable residues is to reduce the complementarity of the tetramerisation interaction surface in PKM2 compared to PKMI, thus favouring PKM2 to be in the inactive dimeric state (this prediction was based on the Adaptive Poisson-Boltzmann Solver [36] to calculate surface potentials of MXE regions S14 Fig and S4 Text for values). We also used mCSM-PPI [37] to assess the effect of the MXE variable residue changes on protein-protein affinity and found most of the residues to be destabilising (See S15 and S16 Figs).

At the same time this allows PKM2 to be dynamically switched to the tetrameric active state depending on a cell's requirements, which could allow cancer cells to more rapidly respond to changes in its environment. The dynamic switching to the active tetramer is provided by an



Fig 5. Examples of MXE events affecting amongst other things. (A) Protein-protein interaction of the PKM gene. Cancer mutated residues are shown in spacefill. (B) Allosteric regulation of the PFKP gene. Cancer mutated residues are shown in spacefill. (C) Small molecule interaction (PSI) sites in the CRMP gene. We are only showing variable binding zinc/catalytic residues. Zinc molecules are shown green spacefill. (D) Catalytic site residues of gene CG42249 from drosophila. We are only showing variable binding zinc/catalytic residues.

https://doi.org/10.1371/journal.pcbi.1008708.g005

allosteric effector, FBP, the binding of which is also provided by the MXE variable residues (discussed below).

Allostery. A total of 391 MXE events were associated with allosteric functions in our analysis. Analysis of PKM MXE structures (along with mutation data [38–40]) identified a key variable residue (Lysine 433) in PKM2 which modulates FBP binding ability (FBP does not bind PKM1). Previous studies have demonstrated that allosteric binding of FBP in PKM2 induces movement/rotations of the helices in the interface and the FBP activating loop that brings the tetramer binding residues into correct alignment for forming the tetramer [41,42]. Therefore, the ability to increase interface complementarity by binding FBP in PKM2 (switching from an inactive dimer to active tetramer) is encoded by the MXE variable residues (Fig 5A).

Interestingly, we find another tightly regulated glycolytic enzyme, that is also allosterically regulated, called Phospho Fructo Kinase-platelet (PFKP) (Fig 5B) and again in which the MXE event likely alters the allosteric regulation (See S17 Fig and S4 Text for detailed analysis).

Furthermore, the MXE regions of PKM1/PKM2 and of PFKP were found to overlap with pan-cancer mutations in structural regions close to functional sites (Fig 5A and 5B), reinforcing the idea that mutations in these regions impact on the function of the proteins and that cancer mutations tend to have similar effects on function as the effects mediated by MXE variable positions.

Small molecule ligand binding. A total of 235 MXE events were associated with proteinsmall molecule interaction functions. An example of this is the Collapsin Response Mediator Protein CRMP gene (FBgn0023023), a dihydropyrimidinase enzyme (EC 3.5.2.2) that has been implicated in biological processes including Notch signalling [43]. An MXE variable residue having zinc-binding function (residue 192), switches from a histidine in isoform CRMP-PA to a glycine in isoform CRMP-PB disrupting the binding of zinc (Fig 5C). Since zinc molecules act as cofactors in enzymes this is likely to reduce enzyme activity for the zinc free isoform. Interestingly, other variable residues in the CRMP MXE produce dramatic changes in the surface charge distribution (S18 Fig). Given the extensive inferred protein interactions for this protein from the IBIS resource (see Materials and Methods), some of which overlap the MXE variable regions, it suggests that MXE switching could also switch CRMP protein interactions.

Catalytic sites. We identify 9 MXE events that affect catalytic sites (CSA residues) in our dataset. For example, CG42249 a poorly functionally characterised fly gene is likely to be a nucleotidase (based on domain assignments and M-CSA conserved catalytic residues (PDB:1USH)). We identify switches in key catalytic residues between its MXE isoforms with major differences in physicochemical properties (Fig 5D). Such changes in physicochemical properties could produce enzymes with very different or even opposing functions and shows the utility of pipeline for identifying interesting candidates for further experimental studies.

PTMs

Phosphorylation. Integrins are an important class of transmembrane proteins in animals mediating cell interactions and signalling in many processes (e.g. embryonic development). We identified an MXE in integrin-B1, where one isoform (ENSMUSP00000119699) has MXE variable residues substituting two residues (Thr788-Thr789 in ENSMUSP0000087457) from Thr to Asn, abolishing a known phosphorylation site. Hence even though the kinase may be expressed and active, only one isoform is capable of recruiting the kinase and so switching isoforms can produce changes in downstream signalling.

Glycosylation. KLRC2 is a transmembrane protein important for regulating Natural Killer (NK) cell-mediated immunity and has an MXE event that switches exon 3, changing 2 amino acids in the extracellular region of the protein. The substitution of Serine 102 to Phenylalanine by this MXE event would remove the glycosylation of amino acid N100 (identified from a 'manual assertion' in UniProt [44] and the N-glycosylation consensus motif). Searching for this MXE in KCLR2 primate orthologs, reveals that the MXE is absent from closely related species such as Orangutan, Bolivian squirrel monkey and Macaques suggesting this a highly dynamic MXE which could help tune the immune response in specific lineages by altering PTM states.

MXE-MOD website

To aid further research into MXE events and guide experimental studies, we developed a webserver called MXE-MOD available at http://gene3d.biochem.ucl.ac.uk/mxemod/. The website displays the MXE events on their modelled structures and shows the locations of functional residues in a simple and intuitive way (Fig 6A). From the website it is possible to browse the list of structurally modelled MXE events and view functional annotations from a number of data sources such as the Gene Ontology database and obtain information on the potential for druggability from the human proteome atlas [31] (Fig 6B).

The website can also be used to identify those isoform pairs in Drosophila, that switch gene expression strongly (i.e. having different RNA-seq expression patterns), via a previously published metric called the Transcript Switch score [45] which can help to identify further biological signatures of interest. In the following section, we highlight two MXE events with high Transcript Switch scores in Drosophila development. The first one alpha-actinin has a role in



Fig 6. MXE-MOD Website. (A) The home page of MXE-MOD website. (B) The browser view allows the user to search different organisms and find a particular gene of interest. Annotations for the gene and associated MXE event have been integrated from different public resources: CATH, GENEONTOLOGY, Human Protein Atlas, DRUGBANK and Ensembl [31,46–49].(C-D) Detailed 'MXE model' pages for 2 different examples in the website. We can see the alternative MXE structure superposed in (C). We can see the variable PPI residues shown in space fill in the website in (D). MXE-MOD utilises the PV javascript viewer [50]. For both examples we show the expression patterns for the MXE isoform groups at different stages of development from RNA-seq and proteomics derived datasets. RNA-seq data is obtained from ModEncode resource [51] and proteomics data from the DDIP consortium (https://ddip-proteome.org) led by Simon Hubbard, Manchester University, UK.

https://doi.org/10.1371/journal.pcbi.1008708.g006

cytoskeleton protein binding and shows strong RNA-seq transcript isoform switching after dorsal closure (the proteome data shows a further lag in time for the switching) (Fig 6C). Clicking the superpose MXE button in the website shows that the alternative MXE has a highly similar structure but the physicochemical features of the variable side chains change (see contrasting gold and blue side chain colours). For this protein we can see there are six variable residues (coloured), two of which are involved in protein interactions of which one undergoes a non-conservative change (S247->C) and would likely alter PPI binding function. In a second example, Tropomyosin-1, a protein involved in actin binding, we see a strong transcript isoform switch around dorsal closure again resulting in remodelling of the PPI interface of the protein (Fig 6D).

Discussion

Homologous mutually exclusive exons are enriched in proteomics data, suggesting a functional role. In this study, we used a large-scale structural modelling analysis to investigate how the homologous exons could be altering function. Previous analyses by other groups have demonstrated that MXE events are evolutionary conserved among vertebrates [17,18] and tend not to disrupt structural domains [17,18]. However, to date, there are no reported largescale, structure-based studies in the literature exploring the likely structural, molecular, mechanistic and functional consequences of homologous MXE switching events that maintain structural integrity. In this work, we identified the homologous MXE events in human, mouse, fugu fish and zebrafish genomes using an in-house computational protocol. Most of these MXE genes are associated with membrane proteins (associated with cell-cell adhesion, signal transduction and molecule/ion transport) and enzymes (e.g. kinases, transferases, hydrolases and oxidoreductases). We annotated about 50% of the MXE events with structural information (i.e. known structures, structural models built using the CATH functional family based Fun-Mod modelling platform and structural mapping) massively increasing the structural annotations available (i.e. by several hundred-fold). We also exploited our in-house CATH functional family resource to identify putative functional sites for proteins involved in MXE events, which significantly increased the amount of functional site data available thereby enabling a much greater, in-depth structure/function analysis than reported to date.

The fact that so many of the 3D modelled domains for MXE events produced reliable structural models (as judged by the structural quality scores such as the normalised DOPE score) suggests that the MXE sequence changes generally do not affect the stable folding of the proteins, but rather tune the functions of the proteins. Although this lack of disruption to the fold has been shown for individual structures previously [16], this is the first time it has been demonstrated using a large-scale structural modelling analysis. Previous structural analyses have also shown that MXE-splicing events tend to lie within structural domains (as opposed to being in e.g. intrinsically disordered regions) [18]. We want to highlight again that all the structures for the analysis are either complete structures from the PDB (though possibly protein subunits in their un-complexed form) or high quality well-packed 3D-models built from a 3D-template which overlaps significantly with the query sequence. Our analysis shows that the MXE events although being within domains, have average surface exposure far above what would be expected by chance. We would expect this phenomenon if the MXE switching events are associated with a change in function, as switching surface exposed residues can induce changes in protein functions (e.g. shifts in protein interfaces and binding pockets, etc.). Some MXE events are not so exposed and may affect buried allosteric sites or destabilise the protein. Furthermore, the variable residues tended to form clusters and generate large changes in physicochemical properties, demonstrating that MXE events can induce functional shifts in proteins.

We examined cancer mutations in an MXE structural context and focused on specific socalled cancer-driver domain families (i.e. MutFams, CATH functional families), which can be statistically associated with cancer mutations. We find a significant proximity (in structure) between cancer mutations in these families and the MXE variable residues, showing that cancer may dysregulate the same functional regions altered by MXE events.

Examination of the recently published MXE-splicing dataset (the Hatje dataset) from human (See S1 Text) showed similar trends. GO annotations enriched in both the Hatje and our main human MXE dataset are similar, both capturing largely membrane associated functions. The Hatje dataset is larger (it uses RNA-seq data to identify further transcripts) and is able to capture GO terms deeper in the hierarchy more easily, but the terms are similar to those for our main dataset at a higher level in the GO hierarchy. Regardless of the MXE dataset used, our structural modelling of the MXE events ensures that only MXEs that occupy equivalent regions of the structure and produce good structures are used in the analysis. Hence our dataset identifies MXE pairs that are homologous and occupy equivalent regions of the protein whilst maintaining the structural integrity of the protein.

Analysis of cassette exons and microexons showed none of the functional enrichments seen for homologous MXE datasets (See S2 and S3 Texts) highlighting a distinct functional role for MXEs. The case of microexons is interesting since they show several indicators of functionality

including being on the surfaces of proteins, and as previously reported, they are under selective pressure to preserve the reading frame [8]. One caveat to our study to be kept in mind is that our analysis will necessarily be biased towards globular structural regions of proteins which are enriched in the PDB. In this analysis, we only analyse exons that map to ordered globular domain structures. MXE regions are more frequently (65%) found in ordered regions than cassette exons (18%). However, that does not exclude other possible functional effects of cassette exons. For example, cassette events that truncate the domain effectively disrupt protein function.

Why MXE events have evolved in some proteins, altering their regulation or functional specificity, to give genes with separate functionalities, remains an open question, but for whatever reason it appears to be particularly important for functional innovations in membrane proteins. Similar mechanisms to those we describe for MXE events can occur in the functional divergence of paralogs [52] i.e. with mutation of residues lying in or near to existing functional sites. It could be argued that MXEs provide a conceptually simpler model for functional switching than would be needed to switch two functionally diverse paralogs, since for the latter, it would be necessary to fully stop transcription of the first paralog and transcribe the second paralog to obtain the alternative function whilst the MXE event is by definition a binary event. However, it must be noted that MXE events are far less frequent than gene duplication events and therefore will have a much smaller impact on the functional repertoire of the organism.

A detailed characterisation of MXE events can help us to better understand cell regulation, various diseases and possibly develop highly specific therapeutics. For example, one reported MXE event that could be targeted therapeutically is the MXE switch between 8a and 8 in the CaV1.2 Calcium Channel [53], which is implicated in multiple diseases and where there appear to be different sensitivities to the same drug, between isoforms. The fact that MXE regions are often in membrane proteins and/or surface exposed identifies them as promising targets for druggable genes. Unlike cassette exons where extensive structural disruption is likely (when occurring in a globular region) the MXE pairs maintain similar overall structures but with altered sidechains, which is ideal for facilitating targeted structure guided drug design to one isoform. Examples of druggable membrane proteins in our dataset includes important targets such as the sodium and calcium voltage-gated channels. Furthermore, there are additional benefits specific to targeting a splice variant. For example, if a drug binds efficiently to its target but the inhibition is too general (e.g. affecting many tissues), unwanted side effects can emerge. However, targeting a minor MXE splice variant could be advantageous since its expression will be much more (e.g. tissue) specific. A splice variant specific drug could target a subset of tissues/functions of the gene whilst leaving the more general function of the gene intact. A promising class of compounds for this are PROTACs which could be specifically engineered to target one splice variant for degradation [54].

A further advantage of MXE splice variants as drug targets is that, as we have demonstrated here, we can obtain reliable structural models for a considerable number of the MXE events, providing a good starting point for rational drug design. Furthermore, since the physicochemical properties of the variable residues show strong shifts between MXE events this should also help in designing isoform specific drugs. Finally, much of the above discussion for therapeutic drug development could equally be applied to developing more targeted insecticides, especially if some of the putative lineage-specific MXEs were validated (e.g. Multidrug-Resistance like Protein 1 in drosophila is one such MXE we identify).

Materials and methods

Identifying MXEs and the amino acid sequence region affected by MXE events

We downloaded the annotations of predicted MXEs for *Drosophila melanogaster* from the Kassiopeia resource [55] which corresponded to FlyBase 5.36 [56] and stored the MXE events and associated protein sequences that we could map to this genome. For example, we did not make use of non-reference genome exons and only used examples where the translations from chromosome sequence exactly matched the FlyBase translation. To help ensure we identified homologous MXE events, we filtered the fly MXEs to only include those with detectable homology (i.e. we compared the sequences against each other using BLAST [57], setting an e-value threshold of 0.005 and a minimum sequence identity of 25%). We also removed MXEs which involved nonsense-mediated mRNA decay transcripts. Note that, the fly DSCAM genes have a total of 4 variable exon clusters giving 38,016 potential splicing isoforms. Because of the dominance of the DSCAM gene in the drosophila dataset we removed it from further analysis so as not to bias the results too heavily towards one protein.

At the time of data preparation, the Kassiopeia resource did not provide MXE download annotations for the other four species (human, mouse, zebra fish and fugu fish). In order to identify MXEs for these species, we did the following. Based on Ensembl version 87 [49], we obtained the mRNA transcripts and identified sets of genes with MXEs. For each coding exon, we identified the longest transcript it was contained in and then compared this against all the other transcripts from the same gene to identify MXEs potential. We made sure that the exons were consecutive in the DNA sequence, but never occurred together in the same transcript and couldn't be linked together in a graph where edges were derived from any overlaps between exons in the gene. We restricted the exons to be similar in length (by checking that the absolute value of the log ratio of the MXE sequence lengths was > 0.25) and that they occurred in the same reading frame. After that, we translated all the MXE exons into amino acid sequences and compared the sequences against each other with BLAST, setting an e-value threshold of 0.005 and minimum sequence identity of 25%. Because we wanted to be more confident of the homology, we filtered out instances where either MXE was < 8 residues. Having at least 8 residues helped us to assess homology by BLAST more easily (i.e. with only a few residues this would not be possible). We allowed terminal exons amongst our MXE dataset and we did not enforce identical flanking exons, but 93% of our events had equivalent flanking exons (precise boundaries of these exons was allowed to vary). The structural modelling step ensures that the MXEs occupy equivalent regions of the structure and are valid structurally. Hence, our dataset focuses on the set of MXE pairs that are homologous and occupy equivalent regions of the protein whilst maintaining the structural integrity, rather than the exon switching mechanism and therefore differs from MXE splicing events defined in the Kassiopeia resource. However, during the preparation of the manuscript, a large and extensive set of MXE-splicing annotations became available for human (human validated-MXEs downloaded from the Kassiopeia website Hs_MXE_validated.gtf). We tested this additional dataset of RNA-seq validated MXE-splicing events (that we refer to as the Hatje dataset). For each of the MXE clusters in this file we identified one exon that mapped to a known coding exon (Using the gff3 file GRCh37.87 from Ensembl) and identified the longest protein sequence this exon was found in. We then picked an alternative exon from the MXE cluster and substituted this for its MXE partner to provide our alternative protein sequence. This step was carried out to be confident that the mapping of exons was done correctly and to obtain the full protein sequence. Furthermore, we sampled only one MXE-pair for each MXE cluster (to prevent biases from large MXE clusters affecting our analysis). Note as part of our pipeline we filtered

the Hatje dataset using the BLAST criterion defined above since we are only reliably able to do the homology model comparisons by enforcing this. The Hatje dataset with RNA-seq and more MXE-splicing splicing filters [18] therefore provides an important comparison as to our dataset. Applying our structural modelling pipeline ensures, in both datasets, the MXE pairs are homologous and occupy equivalent regions of the protein whilst maintaining the structural integrity of the protein.

Orthologs

Orthologs were obtained using Ensembl Compara [58] including all orthology relationship types. This was to make sure we were not being overly conservative, since the main purpose of this step was to check for any potential redundancy (at the ortholog level) between genomes.

Predicting the function of MXE genes

We used the PANTHER functional annotation tool, last updated in February 2020, to identify functional enrichments of the MXE genes. We used 5 different background sets (organism whole genome/ genes that have no paralogs/ multi-exon genes/ multi-protein isoform genes/ MXE or CE genes) for testing the enrichment.

Annotating MXE events with structural information

We mapped the splice isoforms to CATH functional families (FunFams). FunFams are a subclassification of CATH protein domain superfamilies [46]. Each FunFam groups together relatives likely to have very similar structures and functions. They are generated using the FunFHMMer protocol [25] that detects similarities in sequence patterns (highly conserved positions and specificity determining positions). Positions that are highly conserved across a superfamily are generally important for the stability, folding or function of the protein domain. Specificity-determining positions are positions that are conserved within and unique to a particular FunFam, responsible for a specific function and usually involved in functional divergence from other FunFams. The FunFam protocol uses agglomerative clustering to iteratively merge clusters of relatives. This is done by starting with very close homologues (>90% sequence identity) and building multiple sequence alignments (Katoh and Standley, 2013) and Hidden Markov Models (using HMMER [59]) for each cluster. Subsequently all vs all, HMM-HMM comparisons are performed across all the clusters using HHpred [60] and the most similar clusters merged. A new MSA and HMM are built for the merged cluster and another round of HMM-HMM comparisons performed. This is repeated until all clusters have been merged giving a hierarchical tree for the superfamily. Finally, the tree is used to guide the identification of distinct functional family clusters having different specificity determining residues from other clusters in the tree. More comprehensive details of the functional family generation protocol can be found in [25].

The domains within a given FunFam have been demonstrated to be structurally coherent [26,61]. The functional purity of the FunFams has been demonstrated by validating against experimentally determined proteins from the Enzyme Commission and also by checking whether known functional sites coincide with highly conserved residues in the Multiple Sequence Alignments (MSAs) of FunFams [25]. CATH FunFams have been shown to be more functionally pure than Pfam domain families [25,26]. Functional predictions based on Fun-Fams were ranked amongst the top five methods for the "Molecular Function" category and the "Biological Process" category in the most recent CAFA International Function Prediction experiments (CAFA2, [62], CAFA3, [63]).

All the splice events were scanned against the library of CATH v4.1 FunFam HMMs [46,64] using HMMER [59]. DomainFinder3 [65] was used to determine which CATH-Gene3D Fun-Fams they belonged to. We only considered matches with a HMMER E-value of less than 0.001. For FunFams that had a known domain structure, we annotated the splice events using the FlyBase/Ensembl to PDB mapping. For FunFams that had no relative of known structure, we used the FunMod modelling pipeline [61,66] which exploits the MODELLER [67] algorithm to build structural models. We ensured that the sequence to be modelled overlapped with 80% of the residues in the representative sequence for the FunFam. We used normalised DOPE [68] and GA341 [69] to assess the quality of the models. Only good quality models with a negative normalised DOPE score and a GA341 score of more than 0.7 were included in this analysis. For those splice events where we failed to build a model, we mapped them to the structural representative of the respective FunFam. To ensure that we chose a structural representative that represents the FunFam well, the structural domain with the highest cumulative SSAP structural similarity score and the best X-ray resolution was used. SSAP is a well-established structural comparison method [70]. The SSAP score ranges from 0 to 100, with a score of 100 for identical structures.

The two splice isoforms for an MXE event were aligned to other relatives in the FunFam using the multiple sequence alignment tool MAFFT [71] and the function mafft-add. We extracted the alignment between the structural representative of the FunFam and the two splice isoforms. Based on this alignment, we extracted the variable residues between the two isoforms (See Fig 1).

For those MXE events which are not included in FunFams, we scanned the isoforms against the libraries of HMMs built from non-redundant structures in various resources, CATH v4.1 HMMs (i.e. including non-FunFam CATH domains), SCOPe 2.06 HMMs [72] and PDB70 June 2017 HMMs [73]) using either HMMER or HHsearch [60]. Similarly, we only considered matches with an E-value of less than 0.001. We used the FunMod modelling platform to build structural models. For those events where we failed to build a good model, we mapped them to the best structural matches (based either on the HHsearch or the HMMER result). We only analyse MXE events where both isoforms map to the same structural template. We aligned the isoforms with the structural template using MAFFT and extracted the variable residues between the two isoforms.

For those MXE events which we failed to annotate with structure information, we predicted if they were intrinsically disordered using IUPRED [74]. We used the long-disorder option of IUPRED and residues with IUPRED score above 0.5 were considered disordered. We defined a splice MXE isoform as intrinsically disordered if more than 50% of the residues were predicted to be disordered.

Identifying single and minimal sequence loss Cassette exon events

To compare the MXE events with other types of alternative splicing we examined the impacts of single cassette exon events and microexon events, both of which like MXE events would be expected to have a relatively small effect on the protein structure compared to other AS events.

Loss of exons is likely to disrupt protein structure, especially if the loss is in the globular folded region (which is mainly the region that our structural modelling is restricted to). By restricting to single exon events, we hoped to identify cassette exons that were least likely to alter protein structure and should therefore be easier to analyse from a structural point of view. Conversely, we can imagine that in general more extreme splicing events leading to greater losses would be more disruptive to structure and more difficult to analyse from a homology modelling point of view. To minimise this, for each protein coding gene we took

the longest protein isoform and found the next longest isoform where the only difference was the loss of a single coding, non-terminal exon. In this way, we were more likely to identify a minimally disruptive cassette exon event. We used this set of minimally disruptive cassette exons for the domain/ protein function enrichments and the cassette exon structural models. We only analysed cassette exon events that could be mapped to a globular domain in one of our FunFams.

Microexons

Microexons have been defined previously as having ≤ 27 nucelotides (nt). The most comprehensive set of microexons can be found in the VASTDB database [75]. We obtained our microexon sequences by extracting exons with "EX" in the VASTDB identifier (i.e. HsaEX*) and with length ≤ 27 nt (September 2019). Where possible we mapped the nucleotide sequences using microexon and flanking exons to identify corresponding proteins from CDS data in Ensembl genomes allowing us to place the microexon in the full-length protein sequence.

Quantifying the residue changes using a physicochemical score

We compared the physicochemical properties of variable residues using the McLachlan physicochemical similarity matrix [32]. A pair of amino acids was given a similarity score ranging from 0 to 6. A score of 0 indicates no similarity or a deletion. The score for a pair of identical amino acids is typically 5 or 6 [32]. For each MXE, we summed up all the McLachlan similarity scores for a set of variable residues and divided by the total number of mapped variable residues to normalise the score.

Analysis to determine if variable residues in the splice event are exposed to solvent

NACCESS [76] is a stand-alone program that calculates the relative accessible surface (rASA) of amino acids in a PDB structure. For each MXE splice event, we calculated the rASA for all the corresponding variable residues of the two isoforms. Amino acid residues were considered to be exposed if the rASA value was above 10%. We compared the solvent exposure of the variable residues versus the solvent exposure of the MXE conserved residues for each MXE event. The solvent exposure was calculated as:

 $solvent exposure = \frac{Number of exposed residues}{Total number of residues}$

We also investigated random models to examine the solvent exposure for the splice region itself relative to background. For every MXE pair, we randomly selected from the same structure regions that have the same size as the splice region, 10,000 times, and determined the percentage of exposed residues in this region. This percentage is then compared with the splice region solvent exposure. We used the Wilcoxon signed-rank test to compute the statistical significance.

Analysis to determine if variable residues in MXE regions are clustered in 3D and are close to functional sites

We investigated whether exposed variable residues in MXE events lie on or in the vicinity of functional residues. To reduce the noise and identify those variable residues likely to be having similar impacts, we clustered variable residues into structural clusters. For each pair of MXE

events, we calculated the all-atom-versus-all-atom distances of the variable residue atoms of the structural representative and determined the minimum distance between residues. We used an in-house multi-linkage clustering program which clusters residues that are within 8Å distance of each other (i.e. between the centres of the residues). We made sure there are at least three residues present in a cluster.

Subsequently, we calculated the distance from the structural clusters to known functional residues of the structural representative. We calculated the centre-of-mass of the clusters. The minimum distance from the centre-of-mass of the cluster to the closest functional residue was calculated. If the functional residue was in the cluster the distance was simply set to zero. We used a distance cut-off of 6Å to define if the structural cluster lies close to a functional site.

We also investigated random models to examine the proximity of randomly selected residue regions to functional sites, to evaluate statistical significance. For every MXE pair, we randomly selected structural clusters that have the same size as the largest MXE structural cluster, 10,000 times, and determined the percentage of clusters that lie close to functional sites. This was done by randomly selecting a residue and then identifying the residues that are close (less than or equal to 2A) to the selected residue, and each other, until they made up the same volume as the largest MXE cluster. Then, the overall percentage for all random events was computed. This percentage is then compared with the actual MXE percentage. We used the Z-score test to compute the statistical significance.

The functional sites considered were known experimentally characterised sites: catalytic residues taken from CSA [21], protein-protein interaction residues and protein-small molecule interaction residues (small molecule binding) taken from IBIS [77]. We also used our inhouse predicted functional sites, FunSites [25], and predicted allosteric sites. FunSites are highly conserved positions in multiple sequence alignments of the FunFam relatives, not conserved across the whole superfamily. For the prediction of FunSites, we only perform this analysis for FunFams that have a diverse set of relatives making it possible to distinguish conserved positions from variable positions. The Scorecons method [78] was used to calculate the sequence diversity of a FunFam alignment by generating a diversity of positions score (DOPS score). FunFams with a DOPS score of above 70 (out of 100) are deemed sufficiently sequence diverse for analysis (i.e. having a lower probability of predicting false positives for FunSites [26]).

We also used the A-SITE allosteric site predictor (Dr Aurelio Moya Garcia, personal communication), based on the centrality of nodes (in a protein i.e. residue nodes) in terms of their capability to transfer signals through the protein, to predict allosteric sites in the protein, in order to determine whether variable residues lay on or close to allosteric sites. The program identifies residues with high betweenness centrality. The depletion of residues having high betweenness centrality values would be expected to interrupt the allosteric communication among regions of the protein that lie far apart. For each structure, we defined its allosteric residues as the top 5 percentile residues (ranked by the A-SITE program using the betweenness centrality measure). The source code for A-SITE program is available at https://github.com/ amoyag/protein-structure-network.

Analysis to determine if variable regions in human MXE pairs are close to COSMIC cancer mutation residues

COSMIC is a database which collects somatic mutations from human cancer patients [79]. Accumulation of these mutations may alter cellular functions and contribute to cancer [80]. A number of studies have found that cancer mutation sites lie in the vicinity of functional residues (i.e. protein and small molecule binding sites) [33,34].

Recently, mutationally enriched domain functional families (MutFams) have been identified by in-house studies [24]. MutFams are CATH functional families that have been found to be statistically significantly enriched in somatic missense cancer mutations reported in COS-MIC. They are identified using a protocol developed by Miller et al. [81] that searches for significant enrichment of cancer mutations in a specific domain, compared to the protein background, for the set of proteins containing the specific domain family. A total of 541 pancancer MutFams were identified and they indicate important cancer domain families such as P53 and PTEN [24]. MutFam putative driver genes were found to have reasonable overlap with driver genes identified from Cancer Gene Census and genes identified by other methods based on Pfam families. MutFam genes are enriched in survival and cell-motility cancer hallmark processes and additionally identify proteins with G1/S checkpoint function in DNA repair [24]. Pan-cancer mutations clustered on representative domains in MutFams with available structural domains, have been found to be significantly closer to key functional sites than un-clustered cancer or germline disease mutations.

Therefore, we determined if MutFam cancer mutation residues lie close to MXE variable residues. We mapped all the MutFam mutation residues onto the human MXE events. For those where we have structural information, we determined if the MXE variable residue or residue clusters are close to MutFam mutation residues using a distance cut-off of 4Å (between any atoms in the residue pairs) or 6Å (to the centre of the MXE cluster).

Druggable / Tissue gene enrichments

The set of FDA druggable proteins downloaded from the Human Protein Atlas (HPA) and we checked for enrichment against the gene lists from the MXE datasets. The gene background for the Fisher Exact test was set to the set of protein coding genes. The TopAnat enrichments were calculated through the website using the default parameters (node size = 20, all-datasets (gold and silver)) but using only protein coding genes (from GRCh38) as the genomic background and we report the FDR scores from the website, as described in the methods publication [28,29].

Additional tools used to analyse specific examples

We used Adaptive Poisson-Boltzmann Solver (APBS) [36] to determine the surface potentials of two MXE regions. APBS is a software package which models biomolecular solvation by solving the Poisson-Boltzmann equation (a model used to describe electrostatics interactions between solutes in salty, aqueous media). We used mCSM [37] and mCSM-lig [82] to assess the effect of the MXE variable residue changes on protein-protein affinity and protein-small molecule binding affinity. Both methods use graph-based signatures (distance patterns between atoms and are used to represent the protein residue environment) to train a predictive model from a representative dataset.

Supporting information

S1 Fig. MXE statistically enriched CATH superfamilies. For every CATH superfamily identified in each species, we obtained the number of MXE genes mapped to the superfamily, the number of MXE genes that do not map to the superfamily, the number of non-MXE genes mapped to the superfamily and the number of non-MXE genes that do not map to this superfamily. This information was then used to compute a 2x2 contingency table. We then performed the Fisher exact test using the contingency table. We adjusted the p-value using Benjamini-Hochberg correction to account for multiple hypothesis testing. We check the odds ratio is > 1 and p-value < 0.05 for significance. MXE domain families are enriched in

important Metazoan functions such as membrane proteins involved in cell-cell adhesion and signal transduction.

(TIF)

S2 Fig. Enriched biological annotations identified by PANTHER for all MXE genes. We used organism whole genome as background. We were unable to include Fugu fish because PANTHER does not include Fugu fish in their analysis datasets. (TIF)

S3 Fig. Enriched biological annotations identified by PANTHER for MXE genes that have no paralogs. For our background dataset, we only used genes that have no paralogs. Removing the paralogs was to counter potential effects from MXE events being retained after a gene duplication, which could lead to overestimates of functional coherence of the gene set. We were unable to include Fugu fish because PANTHER does not include Fugu fish in their analysis datasets.

(TIF)

S4 Fig. Enriched biological annotations identified by PANTHER for MXE multiexon genes. We used only multi-exon genes as background. Removing single exon genes was to counter potential effects of any functional bias, since by definition our MXE genes require more than one exon. We were unable to include Fugu fish because PANTHER does not include Fugu fish in their analysis datasets. (TIF)

S5 Fig. Enriched biological annotations identified by PANTHER for MXE multi-protein isoform genes. We used only multi-protein isoform genes as background. Removing single protein isoform genes was to counter potential effects of any functional bias. We were unable to include Fugu fish because PANTHER does not include Fugu fish in their analysis datasets. (TIF)

S6 Fig. Enriched biological annotations identified by PANTHER for MXE genes. We used only MXE or CE genes as background to counter potential effects of any functional bias. We were unable to include Fugu fish because PANTHER does not include Fugu fish in their analysis datasets.

(TIF)

S7 Fig. Splice isoforms with structural information. (TIF)

S8 Fig. Physicochemical properties of MXE variable residues versus randomly sampled residues within the same threshold distance of 4Å from a particular class of functional residue (catalytic, ligand binding, PPI). For each FunFam, with high DOPS, and a known structure we took all known functional sites (CSA (catalytic residues), PPI (protein-protein interactions), PSI (protein-small molecule Interactions)) and identified other residues within 4Å from the functional site. We then determined the residue usage and calculated the chemical changes using McLachlan score. We used Mann-Whitney U test to compute the statistical significance.

(TIF)

S9 Fig. Number of structural clusters formed by MXE variable residues. (TIF)

S10 Fig. Proportion of variable region clusters in MXE events that lie close to functional sites compared to the number of random clusters that lie close to functional sites, for the 5 model organisms analysed.

(TIF)

S11 Fig. Proportion of variable residues in MXE events that lie close to functional sites compared to the number of random residues that lie close to functional sites for the 5 model organisms analysed. For every pair of MXE events, we determined if the variable residues within the splice regions lie close to any functional sites. We calculated the minimum distance between the atoms of the residues involved. We used a distance cut-off of 4Å to determine if residues are close. For every pair of MXE events, we created 10000 random models (based on the patterns of location of variable residues (i.e. X-X—X-X, X, variable residues, -, non-variable residues), and determined the percentage of highlighted residues that lie close to functional sites. We used the z-score test to compute the statistical significance. This reports a p-value for the level of significance. There is a statistically significant tendency for MXE events with variable residues to lie close to protein-protein interaction, protein-small molecule and FunSites for both fly and human datasets. There is also a tendency for variable residues to be in the vicinity of allosteric sites for the fly dataset. We had less than 10 events annotated with catalytic sites but these did not usually lie close to a MXE region using a 4Å distance cut-off.

(TIF)

S12 Fig. Enriched biological annotations for cancer-related human MXE genes identified by PANTHER. We used organism whole genome as background. (TIF)

S13 Fig. PKM2 and PKM1. (A) Exon structures of PKM isoforms. (B) Sequences from domain 4 for the 2 isoforms. The splice regions are coloured purple or yellow. (C)The sequence alignment of the splice region. Variable residues are coloured as red. (TIF)

S14 Fig. The splice region and surface potential of PKM1 and PKM2 (with FBP bound). The FBP molecule is coloured in cyan. (TIF)

S15 Fig. Summary of the mCSM predicted changes in stability for MXE residue changes (from PKM1 isoform to PKM2 isoform) located near the protein interface using PKM1 structure (PDB id 3SRF).

(TIF)

S16 Fig. Summary of the mCSM predicted changes in stability for MXE residue changes (from PKM2 isoform to PKM1 isoform) located near the protein interface using PKM2 structure with FBP bound (PDB id 1T5A). (TIF)

S17 Fig. mCSM-lig predicted the ligand binding affinity for FBP binding residue 539 (from valine to isoleucine) to be more stabilising. The analysis was done using PFKP structure (PDB id 4XZ2).

(TIF)

S18 Fig. The change in surface potential between CRMP MXE splice isoforms. Positively charged residues are coloured as blue, negatively charged residues are coloured as red. (TIF)

S19 Fig. General statistics of the Hatje dataset. (A) Distributions of MXE sizes. (B) Sequence identity between MXE pairs. (C) Number of variable residues between MXE pairs. (TIF)

S20 Fig. Enriched biological annotations identified by PANTHER for Hatje MXE genes. We used organism whole genome as background. (TIF)

S21 Fig. Enriched biological annotations identified by PANTHER for Hatje MXE genes that have no paralogs. For our background dataset, we only used genes that have no paralogs. Removing the paralogs was to counter potential effects from MXE events being retained after a gene duplication, which could lead to overestimates of functional coherence of the gene set. (TIF)

S22 Fig. Enriched biological annotations identified by PANTHER for Hatje multiexon genes. We used only multi-exon genes as background. Removing single exon genes was to counter potential effects of any functional bias, since by definition our MXE genes require more than one exon. (TIF)

S23 Fig. Enriched biological annotations identified by PANTHER for Hatje multi-proteinisoform genes. We used only multi-protein-isoform genes as background. Removing single protein-isoform exon genes was to counter potential effects of any functional bias, since by definition our MXE genes require more than one exon. (TIF)

S24 Fig. Enriched biological annotations identified by PANTHER for Hatje genes. We used only MXE or CE genes as background to counter potential effects of any functional bias. (TIF)

S25 Fig. Structural and functional analysis of the Hatje MXE splicing dataset. (A) Proportion of splice isoforms with structural information. (B) Distribution of McLachlan scores for the variable MXE residues. (C) The surface exposure of the MXE events compared to random expectation. (D)The surface exposure of the MXE variable residues compared to the MXE event as a whole. (E) Proximity of MXE variable residue clusters to amino acids assigned to different functional classes (indicated by Icons) CSA = catalytic residues from the Catalytic Site Atlas, PPI = protein interaction sites and PSI = protein-small molecule interactions). (TIF)

S26 Fig. General statistics of cassette exon dataset. (A) Number of cassette exons identified (B) Distribution of cassette exon sizes. (C) Distribution of cassette exon coverage (whole sequence).

(TIF)

S27 Fig. Enriched biological annotations for CE genes identified by PANTHER. We used organism whole genome as background. (TIF)

S28 Fig. Enriched biological annotations for CE genes that have no paralogs. For our background dataset, we only used genes that have no paralogs. Removing the paralogs was to counter potential effects from MXE events being retained after a gene duplication, which

could lead to overestimates of functional coherence of the gene set. (TIF)

S29 Fig. Enriched biological annotations for CE multiexon genes identified by PANTHER. We used only multi-exon genes as background. Removing single exon genes was to counter potential effects of any functional bias, since by definition our MXE genes require more than one exon.

(TIF)

S30 Fig. Percentage of the CATH-Gene3D functional family domain deleted in the shorter isoform.

(TIF)

S31 Fig. Model quality difference between the longer isoform and the shorter isoform. We only show cases where an acceptable quality model can be built for the longer isoform. (TIF)

S32 Fig. nDOPE score difference of CE and MXE structural models. The difference in nDOPE score is given by a Δ nDOPE value (by subtracting the nDOPEs of the two isoforms and then taking the absolute value). (TIF)

S33 Fig. Proportion of deleted regions in CE events that lie close to functional sites, compared to the number of random clusters that lie close to functional sites. (TIF)

S34 Fig. General statistics of MIC dataset. (A) Number of MIC events identified (B) Distributions of MIC size. (TIF)

S35 Fig. Enriched biological annotations for MIC genes identified by PANTHER. We used organism whole genome as background. (TIF)

S36 Fig. Enriched biological annotations for MIC genes that have no paralogs. For our background dataset, we only used genes that have no paralogs. Removing the paralogs was to counter potential effects from MXE events being retained after a gene duplication, which could lead to overestimates of functional coherence of the gene set. (TIF)

S37 Fig. Enriched biological annotations for MIC multiexon genes identified by PAN-THER. We used only multi-exon genes as background. Removing single exon genes was to counter potential effects of any functional bias, since by definition our MXE genes require more than one exon.

(TIF)

S38 Fig. Structural and functional analysis of MIC splicing dataset. (A) Percentage of splice isoforms with structural information. (B) The surface exposure of the MIC residues compared to non-MIC residues.

(TIF)

S39 Fig. Proportion of MIC regions that lie close to functional sites, compared to the number of random clusters that lie close to functional sites. (TIF) S40 Fig. Proportion of neural-regulated MIC regions that lie close to functional sites, compared to the number of random clusters that lie close to functional sites. (TIF)

S1 Text. The following section summarises the results analysing MXE events from the Hatje dataset.

(DOCX)

S2 Text. The following section reports results obtained for the cassette exon analysis. (DOCX)

S3 Text. The following section reports results obtained for the microexon analysis. (DOCX)

S4 Text. The following sections give expanded details on the structural analysis of examples presented in the main text with some more details on methods.

S1 Table. List of MXE genes. (XLSX)

S2 Table. List of druggable MXE targets. (XLSX)

Acknowledgments

The authors thank David Jones for fruitful discussions and Markus Ralser for his helpful comments on the PKM study.

Author Contributions

Conceptualization: Su Datt Lam, Jonathan Lees, Christine A. Orengo.

Data curation: Su Datt Lam, Jonathan Lees.

Formal analysis: Su Datt Lam, M. Madan Babu, Jonathan Lees, Christine A. Orengo.

Funding acquisition: Christine A. Orengo.

Investigation: Su Datt Lam, Jonathan Lees, Christine A. Orengo.

Methodology: Su Datt Lam, Jonathan Lees, Christine A. Orengo.

Project administration: Christine A. Orengo.

Resources: Su Datt Lam, Jonathan Lees.

Software: Su Datt Lam, Jonathan Lees.

Supervision: Christine A. Orengo.

Validation: Jonathan Lees.

Visualization: Su Datt Lam, Jonathan Lees.

Writing - original draft: Su Datt Lam, M. Madan Babu, Jonathan Lees, Christine A. Orengo.

Writing – review & editing: Su Datt Lam, M. Madan Babu, Jonathan Lees, Christine A. Orengo.

References

- Hakim NHA, Majlis BY, Suzuki H, Tsukahara T. Neuron-specific splicing. Biosci Trends. 2017; 11: 16– 22. https://doi.org/10.5582/bst.2016.01169 PMID: 28049883
- Pohl M, Bortfeldt RH, Grützmann K, Schuster S. Alternative splicing of mutually exclusive exons—A review. Biosystems. 2013; 114: 31–38. <u>https://doi.org/10.1016/j.biosystems.2013.07.003</u> PMID: 23850531
- Chan S-N, Low EN Den, Raja Ali RA, Mokhtar NM. Delineating inflammatory bowel disease through transcriptomic studies: current review of progress and evidence. Intest Res. 2018; 16: 374–383. <u>https:// doi.org/10.5217/ir.2018.16.3.374</u> PMID: 30090036
- 4. Chen J, Weiss W a. Alternative splicing in cancer: implications for biology and therapy. Oncogene. 2014; 34: 1–14. https://doi.org/10.1038/onc.2013.570 PMID: 24441040
- Li Y, Sahni N, Pancsa R, McGrail DJ, Xu J, Hua X, et al. Revealing the Determinants of Widespread Alternative Splicing Perturbation in Cancer. Cell Rep. 2017; 21: 798–812. https://doi.org/10.1016/j. celrep.2017.09.071 PMID: 29045845
- Tazi J, Bakkour N, Stamm S. Alternative splicing and disease. Biochim Biophys Acta—Mol Basis Dis. 2009; 1792: 14–26. https://doi.org/10.1016/j.bbadis.2008.09.017 PMID: 18992329
- Buljan M, Chalancon G, Eustermann S, Wagner GP, Fuxreiter M, Bateman A, et al. Tissue-specific splicing of disordered segments that embed binding motifs rewires protein interaction networks. Mol Cell. 2012; 46: 871–883. https://doi.org/10.1016/j.molcel.2012.05.039 PMID: 22749400
- Ellis JD, Barrios-Rodiles M, Çolak R, Irimia M, Kim TH, Calarco JA, et al. Tissue-specific alternative splicing remodels protein-protein interaction networks. Mol Cell. 2012; 46: 884–892. https://doi.org/10. 1016/j.molcel.2012.05.037 PMID: 22749401
- Yang X, Coulombe-Huntington J, Kang S, Sheynkman GM, Hao T, Richardson A, et al. Widespread Expansion of Protein Interaction Capabilities by Alternative Splicing. Cell. 2016; 164: 805–817. https:// doi.org/10.1016/j.cell.2016.01.029 PMID: 26871637
- Baralle FE, Giudice J. Alternative splicing as a regulator of development and tissue identity. Nat Rev Mol Cell Biol. 2017; 18: 437–451. https://doi.org/10.1038/nrm.2017.27 PMID: 28488700
- 11. Cieply B, Carstens RP, Functional roles of alternative splicing factors in human disease. Wiley Interdiscip Rev RNA. 2015; 6: 311–326. https://doi.org/10.1002/wrna.1276 PMID: 25630614
- Gallego-Paez LM, Bordone MC, Leote AC, Saraiva-Agostinho N, Ascensão-Ferreira M, Barbosa-Morais NL. Alternative splicing: the pledge, the turn, and the prestige: The key role of alternative splicing in human biological systems. Hum Genet. 2017; 136: 1015–1042. https://doi.org/10.1007/s00439-017-1790-y PMID: 28374191
- 13. Blencowe BJ. The Relationship between Alternative Splicing and Proteomic Complexity. Trends Biochem Sci. 2017; 42: 407–408. https://doi.org/10.1016/j.tibs.2017.04.001 PMID: 28483376
- Tress ML, Abascal F, Valencia A. Alternative splicing may not be the key to proteome complexity. Trends Biochem Sci. 2016; 42: 98–110. https://doi.org/10.1016/j.tibs.2016.08.008 PMID: 27712956
- Tress ML, Martelli PL, Frankish A, Reeves GA, Wesselink JJ, Yeats C, et al. The implications of alternative splicing in the ENCODE protein complement. Proc Natl Acad Sci. 2007; 104: 5495–5500. https:// doi.org/10.1073/pnas.0700800104 PMID: 17372197
- 16. Abascal F, Ezkurdia I, Rodriguez-Rivas J, Rodriguez JM, del Pozo A, Vázquez J, et al. Alternatively spliced homologous exons have ancient origins and are highly expressed at the protein level. PLoS Comput Biol. 2015; 11: e1004325. https://doi.org/10.1371/journal.pcbi.1004325 PMID: 26061177
- Abascal F, Tress ML, Valencia A. The evolutionary fate of alternatively spliced homologous exons after gene duplication. Genome Biol Evol. 2015; 7: 1392–1403. https://doi.org/10.1093/gbe/evv076 PMID: 25931610
- Hatje K, Rahman R, Vidal RO, Simm D, Hammesfahr B, Bansal V, et al. The landscape of human mutually exclusive splicing. Mol Syst Biol. 2017; 13: 959. https://doi.org/10.15252/msb.20177728 PMID: 29242366
- Hatje K, Kollmar M. Expansion of the mutually exclusive spliced exome in Drosophila. Nat Commun. 2013; 4. https://doi.org/10.1038/ncomms3460 PMID: 24025855
- Soom M, Gessner G, Heuer H, Hoshi T, Heinemann SH. A mutually exclusive alternative exon of slo 1 codes for a neuronal BK channel with altered function. Channels. 2008; 2: 278–282. https://doi.org/10. 4161/chan.2.4.6571 PMID: 18719396
- 21. Furnham N, Holliday GL, De Beer TAP, Jacobsen JOB, Pearson WR, Thornton JM. The Catalytic Site Atlas 2.0: Cataloging catalytic sites and residues identified in enzymes. Nucleic Acids Res. 2014; 42: D485–D489. https://doi.org/10.1093/nar/gkt1243 PMID: 24319146

- Drummond DA, Wilke CO. The evolutionary consequences of erroneous protein synthesis. Nat Rev Genet. 2009; 10: 715–724. https://doi.org/10.1038/nrg2662 PMID: 19763154
- Dawson NL, Sillitoe I, Lees JG, Lam SD, Orengo CA. CATH-Gene3D: Generation of the resource and its use in obtaining structural and functional annotations for protein sequences. Methods in Molecular Biology. 2017. pp. 79–110. https://doi.org/10.1007/978-1-4939-6783-4_4 PMID: 28150234
- Ashford P, Pang CSM, Moya-Garcia AA, Adeyelu T, Orengo C. A {CATH} domain functional family based approach to identify putative cancer driver genes and driver mutations. Sci Rep. 2019; 263. https://doi.org/10.1038/s41598-018-36401-4 PMID: 30670742
- Das S, Lee D, Sillitoe I, Dawson NL, Lees JG, Orengo CA. Functional classification of CATH superfamilies: a domain-based approach for protein function annotation. Bioinformatics. 2015; 31: 3460–3467. https://doi.org/10.1093/bioinformatics/btv398 PMID: 26139634
- Dessailly BH, Dawson NL, Mizuguchi K, Orengo CA. Functional site plasticity in domain superfamilies. Biochim Biophys Acta—Proteins Proteomics. 2013; 1834: 874–889. <u>https://doi.org/10.1016/j.bbapap.</u> 2013.02.042 PMID: 23499848
- 27. Mi H, Muruganujan A, Ebert D, Huang X, Thomas PD. PANTHER version 14: more genomes, a new PANTHER GO-slim and improvements in enrichment analysis tools. Nucleic Acids Res. 2018; 47: D419–D426.
- Bastian F, Parmentier G, Roux J, Moretti S, Laudet V, Robinson-Rechavi M. Bgee: Integrating and comparing heterogeneous transcriptome data among species. Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics). 2008. pp. 124–131. https://doi.org/10.1007/978-3-540-69828-9_12
- Komljenovic A, Roux J, Robinson-Rechavi M, Bastian FB. BgeeDB, an R package for retrieval of curated expression datasets and for gene list expression localization enrichment tests. F1000Research. 2016; 5: 2748. https://doi.org/10.12688/f1000research.9973.2 PMID: 30467516
- Irimia M, Weatheritt RJ, Ellis JD, Parikshak NN, Gonatopoulos-Pournatzis T, Babor M, et al. A Highly Conserved Program of Neuronal Microexons Is Misregulated in Autistic Brains. Cell. 2014; 159: 1511– 1523. https://doi.org/10.1016/j.cell.2014.11.035 PMID: 25525873
- Uhlén M, Fagerberg L, Hallström BM, Lindskog C, Oksvold P, Mardinoglu A, et al. Tissue-based map of the human proteome. Science (80-). 2015; 347: 1260419. <u>https://doi.org/10.1126/science.1260419</u> PMID: 25613900
- 32. McLachlan AD. Repeating sequences and gene duplication in proteins. J Mol Biol. 1972; 64: 417–437. https://doi.org/10.1016/0022-2836(72)90508-6 PMID: 5023183
- David A, Sternberg MJE. The contribution of missense mutations in core and rim residues of protein protein interfaces to human disease. J Mol Biol. 2015; 427: 2886–2898. https://doi.org/10.1016/j.jmb. 2015.07.004 PMID: 26173036
- Yamada KD, Nishi H, Nakata J, Kinoshita K. Structural characterization of single nucleotide variants at ligand binding sites and enzyme active sites of human proteins. Biophys physicobiology. 2016; 13: 157–163. https://doi.org/10.2142/biophysico.13.0_157 PMID: 27924270
- Wong N, Ojo D, Yan J, Tang D. PKM2 contributes to cancer metabolism. Cancer Letters. 2015. pp. 184–191. https://doi.org/10.1016/j.canlet.2014.01.031 PMID: 24508027
- Baker NA, Sept D, Joseph S, Holst MJ, McCammon JA. Electrostatics of nanosystems: application to microtubules and the ribosome. Proc Natl Acad Sci. 2001; 98: 10037–10041. <u>https://doi.org/10.1073/</u> pnas.181342398 PMID: 11517324
- Pires DE V, Ascher DB, Blundell TL. mCSM: predicting the effects of mutations in proteins using graphbased signatures. Bioinformatics. 2014; 30: 335–342. <u>https://doi.org/10.1093/bioinformatics/btt691</u> PMID: 24281696
- Bond CJ, Jurica MS, Mesecar A, Stoddard BL. Determinants of allosteric activation of yeast pyruvate kinase and identification of novel effectors using computational screening. Biochemistry. 2000; 39: 15333–15343. https://doi.org/10.1021/bi001443i PMID: 11112519
- Ikeda Y, Taniguchi N, Noguchi T. Dominant negative role of the glutamic acid residue conserved in the pyruvate kinase M1 isozyme in the heterotropic allosteric effect involving Fructose-1,6-bisphosphate. J Biol Chem. 2000; 275: 9150–9156. https://doi.org/10.1074/jbc.275.13.9150 PMID: 10734049
- Lyssiotis CA, Anastasiou D, Locasale JW, Vander Heiden MG, Christofk HR, Cantley LC. Cellular control mechanisms that regulate pyruvate kinase M2 activity and promote cancer growth. Biomed Res. 2012; 23: 213–217.
- Morgan HP, O'Reilly FJ, Wear MA, O'Neill JR, Fothergill-Gilmore LA, Hupp T, et al. M2 pyruvate kinase provides a mechanism for nutrient sensing and regulation of cell proliferation. Proc Natl Acad Sci. 2013; 110: 5881–5886. https://doi.org/10.1073/pnas.1217157110 PMID: 23530218

- Wang P, Sun C, Zhu T, Xu Y. Structural insight into mechanisms for dynamic regulation of PKM2. Protein Cell. 2015; 6: 275–287. https://doi.org/10.1007/s13238-015-0132-x PMID: 25645022
- 43. Jauffred B, Llense F, Sommer B, Wang Z, Martin C, Bellaiche Y. Regulation of centrosome movements by Numb and the Collapsin Response Mediator Protein during Drosophila sensory progenitor asymmetric division. Development. 2013; 140: 2657–2668. https://doi.org/10.1242/dev.087338 PMID: 23720047
- **44.** The UniProt Consortium. UniProt: a worldwide hub of protein knowledge. Nucleic Acids Res. 2018; 47: D506–D515. https://doi.org/10.1093/nar/gky1049 PMID: 30395287
- Lees JG, Ranea JA, Orengo CA. Identifying and characterising key alternative splicing events in Drosophila development. BMC Genomics. 2015; 16: 608. https://doi.org/10.1186/s12864-015-1674-2 PMID: 26275604
- 46. Dawson NL, Lewis TE, Das S, Lees JG, Lee D, Ashford P, et al. CATH: an expanded resource to predict protein function through structure and sequence. Nucleic Acids Res. 2017; 45: D289–D295. https://doi.org/10.1093/nar/gkw1098 PMID: 27899584
- The Gene Ontology Consortium. Expansion of the Gene Ontology knowledgebase and resources. Nucleic Acids Res. 2017; 45: D331–D338. https://doi.org/10.1093/nar/gkw1108 PMID: 27899567
- Wishart DS, Feunang YD, Guo AC, Lo EJ, Marcu A, Grant JR, et al. DrugBank 5.0: A major update to the DrugBank database for 2018. Nucleic Acids Res. 2018; 46: D1074–D1082. <u>https://doi.org/10.1093/</u> nar/gkx1037 PMID: 29126136
- 49. Zerbino DR, Achuthan P, Akanni W, Amode MR, Barrell D, Bhai J, et al. Ensembl 2018. Nucleic Acids Res. 2018; 46: D754–D761. https://doi.org/10.1093/nar/gkx1098 PMID: 29155950
- 50. Biasini M. pv: v1.8.1. 2015. https://doi.org/10.5281/zenodo.20980
- Celniker SE, Dillon LAL, Gerstein MB, Gunsalus KC, Henikoff S, Karpen GH, et al. Unlocking the secrets of the genome. Nature. 2009; 459: 927–930. https://doi.org/10.1038/459927a PMID: 19536255
- Todd AE, Orengo CA, Thornton JM. Sequence and structural differences between enzyme and nonenzyme homologs. Structure. 2002; 10: 1435–1451. https://doi.org/10.1016/s0969-2126(02)00861-4 PMID: 12377129
- 53. Tang ZZ, Sharma S, Zheng S, Chawla G, Nikolic J, Black DL. Regulation of the mutually exclusive exons 8a and 8 in the CaV1. 2 calcium channel transcript by polypyrimidine tract-binding protein. J Biol Chem. 2011; 286: 10007–10016. https://doi.org/10.1074/jbc.M110.208116 PMID: 21282112
- Stanton BZ, Chory EJ, Crabtree GR. Chemically induced proximity in biology and medicine. Science (80-). 2018; 359: eaao5902. https://doi.org/10.1126/science.aao5902 PMID: 29590011
- 55. Hatje K, Kollmar M. Kassiopeia: a database and web application for the analysis of mutually exclusive exomes of eukaryotes. BMC Genomics. 2014; 15: 115. https://doi.org/10.1186/1471-2164-15-115 PMID: 24507667
- 56. Gramates LS, Marygold SJ, Dos Santos G, Urbano JM, Antonazzo G, Matthews BB, et al. FlyBase at 25: Looking to the future. Nucleic Acids Res. 2017; 45: D663–D671. <u>https://doi.org/10.1093/nar/</u> dkw1016 PMID: 27799470
- Camacho C, Coulouris G, Avagyan V, Ma N, Papadopoulos J, Bealer K, et al. BLAST+: architecture and applications. BMC Bioinformatics. 2009; 10: 421, https://doi.org/10.1186/1471-2105-10-421 PMID: 20003500
- Vilella AJ, Severin J, Ureta-Vidal A, Heng L, Durbin R, Birney E. EnsemblCompara GeneTrees: Complete, duplication-aware phylogenetic trees in vertebrates. Genome Res. 2009; 19: 327–335. https://doi.org/10.1101/gr.073585.107 PMID: 19029536
- Eddy SR. Accelerated profile HMM searches. PLoS Comput Biol. 2011; 7: e1002195. https://doi.org/10. 1371/journal.pcbi.1002195 PMID: 22039361
- Remmert M, Biegert A, Hauser A, Söding J. HHblits: lightning-fast iterative protein sequence searching by HMM-HMM alignment. Nat Methods. 2012; 9: 173–175. <u>https://doi.org/10.1038/nmeth.1818</u> PMID: 22198341
- Lam SD, Das S, Sillitoe I, Orengo C. An overview of comparative modelling and resources dedicated to large-scale modelling of genome sequences. Acta Crystallogr Sect D. 2017; 73: 628–640. <u>https://doi.org/10.1107/S2059798317008920</u> PMID: 28777078
- **62.** Jiang Y, Oron TR, Clark WT, Bankapur AR, D'Andrea D, Lepore R, et al. An expanded evaluation of protein function prediction methods shows an improvement in accuracy. Genome Biol. 2016; 17: 1–17. https://doi.org/10.1186/s13059-015-0866-z PMID: 26753840
- 63. Zhou N, Jiang Y, Bergquist TR, Lee AJ, Kacsoh BZ, Crocker AW, et al. The CAFA challenge reports improved protein function prediction and new functional annotations for hundreds of genes through experimental screens. Genome Biol. 2019; 20: 244. <u>https://doi.org/10.1186/s13059-019-1835-8</u> PMID: 31744546

- Lewis TE, Sillitoe I, Dawson N, Lam SD, Clarke T, Lee D, et al. Gene3D: Extensive prediction of globular domains in proteins. Nucleic Acids Res. 2018; 46: D435–D439. <u>https://doi.org/10.1093/nar/gkx1069</u> PMID: 29112716
- Yeats C, Redfern OC, Orengo C. A fast and automated solution for accurately resolving protein domain architectures. Bioinformatics. 2010; 26: 745–751. https://doi.org/10.1093/bioinformatics/btq034 PMID: 20118117
- Lam SD, Dawson NL, Das S, Sillitoe I, Ashford P, Lee D, et al. Gene3D: expanding the utility of domain assignments. Nucleic Acids Res. 2016; 44: D404–D409. https://doi.org/10.1093/nar/gkv1231 PMID: 26578585
- Sali A, Blundell TL. Comparative protein modelling by satisfaction of spatial restraints. J Mol Biol. 1993; 64: 779–815. https://doi.org/10.1006/jmbi.1993.1626 PMID: 8254673
- 68. Shen M, Sali A. Statistical potential for assessment and prediction of protein structures. Protein Sci. 2006; 15: 2507–2524. https://doi.org/10.1110/ps.062416606 PMID: 17075131
- Melo F, Sánchez R, Sali A. Statistical potentials for fold assessment. Protein Sci. 2002; 11: 430–448. https://doi.org/10.1002/pro.110430 PMID: 11790853
- 70. Taylor WR, Orengo CA. Protein structure alignment. J Mol Biol. 1989; 208: 1–22. https://doi.org/10. 1016/0022-2836(89)90084-3 PMID: 2769748
- Katoh K, Standley DM. MAFFT multiple sequence alignment software version 7: improvements in performance and usability. Mol Biol Evol. 2013; 30: 772–780. https://doi.org/10.1093/molbev/mst010 PMID: 23329690
- 72. Fox NK, Brenner SE, Chandonia J-M. SCOPe: Structural Classification of Proteins—extended, integrating SCOP and ASTRAL data and classification of new structures. Nucleic Acids Res. 2014; 42: D304– D309. https://doi.org/10.1093/nar/gkt1240 PMID: 24304899
- 73. Berman HM, Westbrook J, Feng Z, Gilliland G, Bhat TN, Weissig H, et al. The Protein Data Bank. Nucleic Acids Res. 2000; 28: 235–242. https://doi.org/10.1093/nar/28.1.235 PMID: 10592235
- Dosztanyi Z, Csizmok V, Tompa PP, Simon II, Dosztányi Z, Csizmók V, et al. The pairwise energy content estimated from amino acid composition discriminates between folded and intrinsically unstructured proteins. J Mol Biol. 2005; 347: 827–839. https://doi.org/10.1016/j.jmb.2005.01.071 PMID: 15769473
- 75. Tapial J, Ha KCH, Sterne-Weiler T, Gohr A, Braunschweig U, Hermoso-Pulido A, et al. An atlas of alternative splicing profiles and functional associations reveals new regulatory programs and genes that simultaneously express multiple major isoforms. Genome Res. 2017; 27: 1759–1768. Available: http:// genome.cshlp.org/content/27/10/1759.abstract https://doi.org/10.1101/gr.220962.117 PMID: 28855263
- 76. Hubbard SJ, Thornton JM. Naccess. Computer Program, Department of Biochemistry and Molecular Biology, University College London. 1993.
- Shoemaker BA, Zhang D, Tyagi M, Thangudu RR, Fong JH, Marchler-Bauer A, et al. IBIS (Inferred Biomolecular Interaction Server) reports, predicts and integrates multiple types of conserved interactions for proteins. Nucleic Acids Res. 2012; 40: D834–D840. https://doi.org/10.1093/nar/gkr997 PMID: 22102591
- 78. Valdar WSJ. Scoring residue conservation. Proteins Struct Funct Bioinforma. 2002; 48: 227–241. https://doi.org/10.1002/prot.10146 PMID: 12112692
- 79. Forbes SA, Beare D, Boutselakis H, Bamford S, Bindal N, Tate J, et al. COSMIC: somatic cancer genetics at high-resolution. Nucleic Acids Res. 2016/11/28. 2017; 45: D777–D783. https://doi.org/10.1093/ nar/gkw1121 PMID: 27899578
- 80. Martincorena I, Campbell PJ. Somatic mutation in cancer and normal cells. Science (80-). 2015; 349: 1483–1489. https://doi.org/10.1126/science.aab4082 PMID: 26404825
- Miller ML, Reznik E, Gauthier NP, Aksoy BA, Korkut A, Gao J, et al. Pan-Cancer Analysis of Mutation Hotspots in Protein Domains. Cell Syst. 2015; 1: 197–209. <u>https://doi.org/10.1016/j.cels.2015.08.014</u> PMID: 27135912
- Pires DEV, Blundell TL, Ascher DB. MCSM-lig: Quantifying the effects of mutations on protein-small molecule affinity in genetic disease and emergence of drug resistance. Sci Rep. 2016; 6: 29575. https:// doi.org/10.1038/srep29575 PMID: 27384129